Author's response to reviews

Title: The burden of premature mortality in Spain using standard expected years of life lost: a population-based study

Authors:

Ricard Gènova-Maleras (ricard.genova@salud.madrid.org)
Ferrán Catalá-López (ferran_catala@hotmail.com)
Nerea Fernández de Larrea-Baz (nerea.fernandez@salud.madrid.org)
Elena Álvarez-Martín (elena.alvarez.martin@urjc.es)
Consuelo Morant-Ginestar (consuelo.morant@salud.madrid.org)

Version: 2 Date: 5 October 2011

Author's response to reviews:

Valencia, 5th October 2011

Dear Editors and BMC Editorial Team,

Further to your emails and following your publication procedures, we confirm that we have revised the current version of the manuscript and made some minor format corrections. Please, find upload our revised version.

Yours faithfully

Ferrán Catalá-López